Neurobiology of Disease

M2 Cortex Circuitry and Sensory-Induced Behavioral Alterations in Huntington's Disease: Role of Superior Colliculus

Sara Conde-Berriozabal,^{1,2,3} Lia García-Gilabert,^{1,2,3} Esther García-García,^{1,2,3} Laia Sitjà-Roqueta,^{1,2,3} Xavier López-Gil,^{2,4} Emma Muñoz-Moreno,^{2,4} Mehdi Boutagouga Boudjadja,⁵ Guadalupe Soria,^{2,4,6,7} Manuel J Rodríguez,^{1,2,3} Jordi Alberch,^{1,2,3} and ⁽ⁱ⁾Mercè Masana^{1,2,3}

¹Department of Biomedical Sciences, Institute of Neurosciences, School of Medicine and Health Sciences, University of Barcelona, 08036, Barcelona, Spain, ²Institut d'Investigacions Biomèdiques August Pi i Sunyer, Barcelona, 08036, Spain, ³Centro de Investigación Biomédica en Red sobre Enfermedades Neurodegenerativas, Madrid, 28029, Spain, ⁴Magnetic Resonance Imaging Core Facility, Institut d'Investigacions Biomèdiques August Pi i Sunyer, Barcelona, 08036, Spain, ⁵Metofico Ltd, London, WC2H 9JQ, United Kingdom, ⁶Laboratory of Surgical Neuroanatomy, Institute of Neurosciences, School of Medicine and Health Sciences, Universitat de Barcelona, 08036, Barcelona, Spain, and ⁷Centro de Investigación Biomédica en Red de Bioingeniería, Biomateriales y Nanomedicina, Madrid, 28029, Spain

Early and progressive cortico-striatal circuit alterations have been widely characterized in Huntington's disease (HD) patients. Cortical premotor area, M2 cortex in rodents, is the most affected cortical input to the striatum from early stages in patients and is associated to the motor learning deficits present in HD mice. Yet, M2 cortex sends additional long-range axon collaterals to diverse output brain regions beyond basal ganglia. Here, we aimed to elucidate the contribution of M2 cortex projections to HD pathophysiology in mice. Using fMRI, M2 cortex showed most prominent functional connectivity alterations with the superior colliculus (SC) in symptomatic R6/1 HD male mice. Structural alterations were also detected by tractography, although diffusion weighted imaging measurements suggested preserved SC structure and similar electrophysiological responses were obtained in the SC on optogenetic stimulation of M2 cortical axons. Male and female HD mice showed behavioral alterations linked to SC function, including decreased defensive behavioral responses toward unexpected stimuli, such as a moving robo-beetle, and decreased locomotion on an unexpected flash of light. Additionally, GCamp6f fluorescence recordings with fiber photometry showed that M2 cortex activity was engaged by the presence of a randomly moving robo-bettle, an effect absent in HD male mice. Moreover, acute chemogenetic M2 cortex inhibition in WT mice shift behavioral responses toward an HD phenotype. Collectively, our findings highlight the involvement of M2 cortex activity in visual stimuli-induced behavioral responses, which are deeply altered in the R6/1 HD mouse model.

Key words: approach/defensive behavior; fiber photometry; mouse fMRI; movement disorders; optogenetics; premotor cortex.

Received June 15, 2022; revised Feb. 1, 2023; accepted Mar. 1, 2023.

Author contributions: S.C.-B., G.S., M.J.R., J.A., and M.M. designed research; S.C.-B., L.G.-G., E.G.-G., L.S.-R., X.L.-G., and E.M.-M. performed research; S.C.-B., L.G.-G., E.G.-G., L.S.-R., E.M.-M., M.B.-B., G.S., M.J.R., and M.M. analyzed data; S.C.-B. and M.M. wrote the first draft of the paper; S.C.-B., L.G.-G., E.G.-G., L.S.-R., X.L.-G., E.M.-M., M.B.-B., G.S., M.J.R., J.A., and M.M. edited the paper; M.B.-B., J.A., and M.M. contributed unpublished reagents/analytic tools; M.M. wrote the paper.

This research is part of NEUROPA. The NEUROPA Project has received funding from the European Union's Horizon 2020 Research and Innovation Program under Grant Agreement 863214 to M.M. This work was supported by the Spanish Ministry of Sciences, Innovation and Universities under Project PID2021-1248960A-100 (to M.M.) and PID2020-119386RB-100 (to J.A. and M.J.R.); Instituto de Salud Carlos III, Ministerio de Ciencia, Innovación y Universidades and European Regional Development Fund (Centro de Investigación Biomédica en Red sobre Enfermedades Neurodegenerativas, to J.A.); and María de Maeztu Unit of Excellence, Institute of Neurosciences, University of Barcelona, MDM-2017-0729, Ministry of Science, Innovation, and Universities. We thank Ana López, Maria Teresa Muñoz, and Albert Coll for excellent technical support; the staff of the Confocal Microscopy Service of the Scientific and Technological Centers of the University of Barcelona; and the Magnetic Resonance Imaging Core Facility of the Institut d'Investigacions Biomèdiques August Pi i Sunyer for the scientific technical support in MRI acquisition and analysis.

M.B.B. is affiliated with Metofico Ltd. Metofico is the provider of the software used for fiber photometry data analysis. The remaining authors declare no competing financial interests.

Correspondence should be addressed to Mercè Masana at mmasana@ub.edu.

https://doi.org/10.1523/JNEUROSCI.1172-22.2023

Copyright © 2023 Conde-Berriozabal et al.

This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International license, which permits unrestricted use, distribution and reproduction in any medium provided that the original work is properly attributed.

Significance Statement

Understanding brain circuit alterations in brain disorders is critical for developing circuit-based therapeutic interventions. The cortico-striatal circuit is the most prominently disturbed in Huntington's disease (HD); and particularly, M2 cortex has a prominent role. However, the same M2 cortical neurons send additional projections to several brain regions beyond striatum. We characterized new structural and functional circuitry alterations of M2 cortex in HD mouse models and found that M2 cortex projection to the superior colliculus (SC) was deeply impaired. Moreover, we describe differential responses to unexpected sensory stimulus in HD mouse models, which relies on SC function. Our data highlight the involvement of M2 cortex in SC-dependent sensory processing and its alterations in HD pathophysiology.

Introduction

Huntington's disease (HD) is a neurodegenerative disorder characterized by devastating motor symptoms, including chorea, and preceded by cognitive and psychiatric disturbances. HD pathology is caused by a CAG trinucleotide expansion (>36 repeats) in the huntingtin gene (Huntington's Disease Collaborative Research Group, 1993) and results in a progressive degeneration of medium-sized spiny neurons in the caudate and putamen (striatum). The cortex is the main input to the striatum, and an early and progressive cortico-striatal disconnection has been extensively demonstrated by several studies in HD patients (Unschuld et al., 2012; Dogan et al., 2015) and animal models (Cepeda et al., 2007; Veldman and Yang, 2018). In HD patients, the most affected structural and functional striatal connections arise from cortical premotor areas, analogous to the secondary motor (M2) cortex in rodents. Notably, these affectations appear many years before the onset of motor symptoms in HD carriers (Unschuld et al., 2012; Shaffer et al., 2017; Estevez-Fraga et al., 2021; Johnson et al., 2021) and are profoundly impaired in animal models (Hintiryan et al., 2016; Creus-Muncunill et al., 2019; Fernández-García et al., 2020). However, it is unclear whether dysregulated information flow from the M2 cortex in HD affects only cortico-striatal functions or could have an additional impact in the remaining long-distance output nuclei.

Cortico-striatal circuits are required for motor learning processes (Costa et al., 2004; Cao et al., 2015; Kupferschmidt et al., 2017), and M2 cortex projection to the striatum has been associated to the motor learning deficits that characterize HD mice (Fernández-García et al., 2020). Furthermore, M2 cortex is involved in a variety of functions other than motor learning (Cao et al., 2015), such as decision-making (Sul et al., 2011) and perceptual behavior (Duan et al., 2021), as widely reviewed (Barthas and Kwan, 2017; Yang and Kwan, 2021). In this regard, pyramidal tract neurons from M2 cortex project ipsilaterally to striatum and also send collateral axons to the thalamus, subthalamic nucleus, superior colliculus (SC), pons, and spinal cord (Zhang et al., 2016; Economo et al., 2018; Gerfen et al., 2018). Yet, the involvement of M2 cortex circuitry to HD pathology has not been explored beyond basal ganglia.

In the present work, we aimed to determine the contribution of additional M2 cortex circuits to HD pathology using the *R6/1* HD mouse model. First, we mapped alterations in M2 cortex functional connectivity with the rest of the brain in WT and HD mice, by using *in vivo* resting-state fMRI techniques. Then, we selected the SC as key M2 cortex target brain region and studied the structural connectivity of M2 cortex–SC pathway by diffusion MRI analysis. To determine M2 cortex–SC functional alterations in HD mice, we took advantage of optogenetic tools, fiber photometry, chemogenetics, and SC-dependent behavioral responses to unexpected visual stimuli, such as the presence of a moving robo-beetle (Almada et al., 2018) and an unexpected flash of light (Liang et al., 2015). Together, our results show the involvement of M2 cortex activity in visual stimuli-induced behavioral responses and that the M2 cortex activity is deeply impaired in the R6/1 HD mouse model.

Materials and Methods

Animals

Experiments were performed using R6/1 transgenic mice (B6CBA background; RRID:IMSR_JAX:002809) expressing the exon 1 of *mutant huntingtin* with ~145 CAG repeats at the time of experiments and their age-matched WT littermate as controls. Mice were acquired from The Jackson Laboratory and housed together in groups of mixed genotypes with access to food and water *ad libitum* in a colony room kept at 19°C-22°C and 40%-60% humidity, under a 12:12 h light/dark cycle. Genotypes were determined by PCR from ear biopsy, mice randomly assigned to experimental groups, and data were recorded for analysis by microchip mouse number.

Male and female mice were tested at 12 weeks of age for presymptomatic behavioral screening; male mice were tested in experiments during symptomatic phase (~20-week-old). When required, stereotaxic surgeries were performed 4 weeks before testing (16-week-old mice). A power analysis was performed for fMRI and the behavioral experiments based on previous data from the laboratory, considering power of 0.8, α of 0.05, and SD of 20%, which leads to N= 12 for behavior. Nvalues (biological replicates) are given throughout the manuscript in the figure legends. Each experiment was performed once. All animal procedures were conducted in accordance with the Spanish RD 53/ 2013 and European 2010/63/UE regulations for the care and use of laboratory animals and approved by the animal experimentation Ethics Committee of the Universitat de Barcelona and Generalitat de Catalunya.

MRI acquisition

Experiments were blindly performed on a 7.0T BioSpec 70/30 horizontal animal scanner (Bruker BioSpin), equipped with an actively covered gradient structure (400 mT/m, 12 cm inner diameter). To evaluate connectivity between ROIs, mice (n = 11 WT, n = 13 R6/1) at 17-20 weeks of age underwent structural T2-weighted imaging, diffusion weighted imaging (DWI), and rs-fMRI. Animals were placed in a Plexiglas holder in a supine position and were fixed using tooth and ear bars and adhesive tape; a combination of anesthetic gases [medetomidine (bolus of 0.3 mg/kg, 0.6 mg/kg/h infusion) and isoflurane (0.5%)] was administered using a Plexiglas holder with a nose cone.

Proper position of the head at the isocenter of the magnet was ensured with a 3D-localizer scan. T2-weighted image was obtained using a RARE sequence [effective TE = 33 ms, TR = 2.3 s, RARE factor = 8, voxel size = $0.08 \times 0.08 \text{ mm}^2$, slice thickness = 0.5 mm]. DWI was acquired using an EPI sequence [TR = 6s, TE = 27.7 ms, voxel size $0.21 \times 0.21 \text{ mm}^2$, slice thickness = 0.5 mm, 30 gradient directions with b = 1000 s/mm^2 and 5 baseline images without diffusion weighting]. rs-fMRI acquisition was performed using an EPI sequence [TR = 2 s, TE = 19.4 ms, voxel size $0.21 \times 0.21 \text{ mm}^2$,



Figure 1. Functional connectivity analysis using rs-fMRI indicates the SC as the brain region most severely disconnected from M2 cortex in 20-week-old HD mice. *A*, Anatomical regions for functional connectivity analysis based on atlas automatic parcellation. *B*, Average functional connectivity maps of left M2 cortex in WT and the *R6/1* mouse model of HD. Color maps represent the average correlation value when >0.15. *C*, Average functional connectivity of the left M2 cortex with all brain areas analyzed from the left hemisphere are represented. Average of the seed-based correlation map in each area represents functional connectivity between the M2 cortex and each region. Each point represents data from an individual mouse. Two-way ANOVA with genotype and brain region as factors was performed, followed by Bonferroni *post hoc* comparisons test. Data are mean \pm SEM (WT n = 11 and HD n = 13 mice; 17-week-old mice). *p < 0.05, **p < 0.01, ***p < 0.001, HD versus WT. A, Amygdala; Acd, nucleus accumbens; BNST, bed nucleus of the stria terminalis; Cg, cingulate cortex; Ent, entorhinal cortex; Pir, piriform cortex; AI, agranular insular cortex; GP, globus pallidus; Hip, hippocampus; H, hypothalamus; MM, medial mammillary nucleus; SN, substantia nigra; Me, mesencephalic nucleus; mPFC, medial prefrontal cortex; OB, olfactory bulb; OFC, orbitofrontal cortex; PAG, periaqueductal gray; PA, preoptic area; M1, primary motor cortex; M2, secondary motor cortex; RS, retrosplenial cortex; Som, secondary somatosensory cortex; Str, striatum; Thal, thalamus; VP, ventral pallidum.

slice thickness = 0.5 mm]; 420 volumes were acquired and the total scan time was 14 min. After completion of the imaging session, atipamezol (Antisedan, Pfizer) and saline were injected to reverse the sedative effect and compensate fluid loss.

Functional connectivity analysis

Seed-based analysis was performed as described previously (Fernández-García et al., 2020) to evaluate functional connectivity of the M2 cortex with the rest of the brain. Preprocessing of the rs-fMRI data included: slice timing, spatial realignment using SPM8 for motion correction, elastic registration to the T2-weighted volume using ANTs for EPI distortion correction (Avants et al., 2008), detrend, smoothing (FWHM = 0.6 mm), frequency filtering of time series between 0.01 and 0.1 Hz, and regression by motion parameters using NiTime (http:// nipy.org/nitime).

Brain parcellation was defined according to the MRI-based atlas of the mouse brain (Ma et al., 2008). The sensorimotor cortex defined in the original atlas was manually divided into somatosensory cortex (Som), orbitofrontal cortex (OFC), primary motor (M1) cortex, M2 cortex, cingulate cortex (Cg), mPFC, and retrosplenial cortex (RS). The atlas template was elastically registered to each subject T2weighted volume using ANTs (Avants et al., 2008) and finally applied to the label map to obtain specific parcellation for each animal.

Brain parcellation was registered from T2weighted to resting-state fMRI images. The left M2 cortex was selected as seed region for seedbased analysis. Extracted average time series in the M2 cortex was correlated with the time series of all the voxels within the brain, resulting in a connectivity map containing the value of the correlation between the BOLD signal time series in each voxel with the seed time series. Seed to region connectivity was calculated as the mean value of the correlation map in each region, considering only positive correlations.

Structural connectivity analysis

The fiber tracts connecting M2 cortex with SC were identified and characterized based on automatically identified regions in the T2weighted image and tractography results from DWI. Diffusion tensor images (DTI) were estimated, and standard DTI metrics were computed, including fractional anisotropy (FA), mean diffusivity (MD), axial diffusivity (AD), and radial diffusivity (RD). To estimate the fiber tracts, DWI was preprocessed, including eddy-current correction using FSL (Jenkinson et al., 2012), denoising (Coupe et al., 2008), and bias correction (Tustison et al., 2010). The five baseline images were averaged and registered to the T2-weighted image to correct for EPI distortions. Whole-brain tractography was performed using a deterministic algorithm based on constrained spherical deconvolution model, considering as seed points the voxels where FA > 0.1. The same threshold was defined as stop criterion for the algorithm. Once the fiber tracts were identified, its average FA, MD, RD, and AD were computed.

Stereotaxic surgery

We used different adeno-associated virus (AAV): AAV-ChR2 virus construct under CamKII promotor (AAV1-CaMKIIa-hChR2 (H134H)-eYFP-WPRE.hGH; catalog #AV-1-26969P, University of Pennsylvania-Penn Vector core) was used for optogenetic modulation; AAV-GCaMP6f virus construct under SYN promotor (AAV9-Syn-GCaMP6f-WPRE-SV40; Addgene catalog #100837-AAV9) was used for fiber photometry analysis. AAV-hM3D(Gi) virus under hSyn promotor (AAV5-hSyn-hM3D (Gi)-mCherry; catalog #50475-AAV5) was used for chemogenetic inhibition. Virus production, amplification, and purification were performed by University of Pennsylvania-Penn Vector Core and Addgene (titers: ${\sim}1$ ${\times}$ 10¹² genomic particles/ml).

Stereotaxic surgery was performed under isoflurane anesthesia (5% induction and 1.5% maintenance) in 16-week-old mice. Meloxicam (2 mg/kg s.c.) was injected before the surgery to avoid pain and reduce inflammation. A volume of $0.5 \,\mu$ l of corresponding viral constructs was injected in M2 cortex (mm from bregma and dura matter: 2.46 AP,

 ± 1 ML, -0.8 DV) using 5 μ l Hamilton syringe with a 33 gauge needle at 0.1 μ l/min. To allow diffusion of virus particles and avoid refluxes, the needle was left for an additional 5 min period. Animals were housed for viral expression and recovery from surgery at least 4 weeks before behavioral experiments were initiated.

For fiber photometry analysis *in vivo*, fiber-optic cannulas (MFC_400/ 430-0.66_1.0 mm_MF1.25_FLT; Doric Lenses) were implanted unilaterally in the left hemisphere of the M2 cortex (mm from bregma and dura matter: 2.46 AP, ± 1 ML, -0.8 DV). All fiber-optic cannulas were implanted during surgery and fixed in place using dental cement.

Behavioral assessment

All behavioral tests were performed by an experimented observer during the light phase. Animals were habituated to the experimental room for at least 1 h before testing. All testing apparatus were cleaned with water and dried between tests and animals.

Beetle Mania Task (BMT). The BMT assesses both passive and, in particular, active fear responses of the mice to an unexpected moving beetle (Nano Nitro, Hexbug) (Heinz et al., 2017; Almada et al., 2018). The test was performed using a white rectangular arena (15 imes 40 imes30 cm) with dim light (\sim 20 lux). Briefly, the test comprises two successive phases of 5 min: during the habituation phase, mice freely explored the arena. During this time, distance traveled and rearing times were scored using the SMART 3.0 software (Panlab). In the testing phase, confrontations with the erratically moving robo-beetle were scored. The robobeetle was introduced to the arena at maximal distance to the mouse and the following behavioral responses analyzed: (1) escape, mice quickly jump (flight) with accelerated speed in the direction opposite to the beetle's movement vector; (2) tolerance, ignorance of the robo-beetle after physical contact; (3) approach, mice follow the robo-beetle in close vicinity; (4) avoidance, mouse walks in opposite direction to the robo-beetle; and (5) total responses, the sum of all the events evaluated. Escape events were normalized to total responses, whereas tolerance, approach, and avoidance were normalized to the sum of these three parameters, without the escape events, as previously described (Almada et al., 2018).

Light-induced locomotion. The task assesses light-triggered behaviors in freely moving mice and was performed based on Liang et al. (2015). The arena consisted of a narrow and long hall ($8 \times 150 \times 30 \text{ cm}^3$) dimly illuminated ($\sim 20 \text{ lux}$) with a bulb with bright and white light placed at equity distance from the beginning and the end of the corridor. Mice were placed in one site of the hall and freely walked to the opposite site; then, when passed below the bulb, it triggered an ~ 1 s light flash above



Figure 2. Diffusion metrics indicate structural alterations in M2 cortex–SC tracts of HD mice. **A**, Representative tracts connecting the M2 cortex area with the SC in WT and the *R6/1* mouse model of HD. M2 cortex and SC areas were automatically delimited according to atlas. **B**, **E**, Diffusion metrics analysis of tracts between M2 cortex and SC includes (**B**) FA, (**C**) MD, (**D**) AD, and (**E**) RD. Each point represents data from an individual mouse. Unpaired Student's *t* test was performed. Data are mean \pm SEM (WT *n* = 11 and HD *n* = 13 mice; ~20-week-old mice). ***p* < 0.01.



Figure 3. Regional diffusion metrics indicate preserved microstructure in the SC of HD mice. *A*, FA, (*B*) MD, (*C*) AD, and (*D*) RD were measured in the SC of 20-week-old WT and *R6/1* mice. Each point represents data from an individual mouse. Unpaired Student's *t* test was performed. Data are mean \pm SEM (WT *n* = 11 and HD *n* = 13 mice; ~20-week-old mice).

the animal. Each animal was tested for 5 trials, with intertrial intervals of \sim 20 min, and scored: Time 1, time mice walked from the beginning of the hall to the bulb; and Time 2, time mice walked from the bulb to the end of the corridor.

Fiber photometry

Changes of neuronal activity were assessed using the GCamp6f fluorescent calcium indicator and normalized using isosbestic fluorescent recordings (405 nm), using a custom-made fiber photometry system obtained from Doric Lenses. We used the free Doric Neuroscience studio Software to control the Doric console and LED drivers. A 465 nm LED (CLED_465, Doric Lenses) modulated at 241.58 MHz and a 405_nm LED



Figure 4. Optogenetic stimulation of M2 cortex terminals in the SC evoked similar electrophysiological responses in WT and HD mice. *A*, Schematic representation of AAV-CamKII-ChR2-YFP construct injection at M2 cortex. *B*, Representative MEA location in coronal slices containing the SC with axons from M2 cortex expressing ChR2. *C*, Amplitude of fPSC triggered by increasing light intensities in the SC of WT and HD mice. Data are mean \pm SEM (WT n = 7 and HD n = 8 mice; 21-week-old mice).

(CLED_405, Doric Lenses) modulated at 572.21 MHz were directed and coupled into a personalized fluorescence mini cube (iFMC7, Doric Lenses) through 1 m attenuator fiber-optic patchcords ($400 \mu m$ core, 0.48 NA). Combined 465_nm and 405_nm light was launched to a pigtailed ($400 \mu m$ core, 0.57 NA) fiber-optic rotary joint (Doric Lenses) connected to a low autofluorescence mono fiber-optic patchcord ($400 \mu m$ core, 0.57 NA) allowing freely moving tests with mice, and finally mated to the implanted fiber-optic cannula (see stereotaxic surgery) via a black covered Zicronia sleeve. Emitted fluorescence travels back through the mini cube and spectra, detected and amplified by the Fluorescence Detector Amplifier from Doric. Data were sent to the fiber photometry console. Fiberoptic patchcords were bleached for 3 h before each experiment. Mice were connected to the optical fiber 5 min before the task and the 465 and 405 LED were turned on to ensure stable photometric recordings.

Raw and demodulated 465 nm and 405 nm fluorescent intensity data were recorded at 12,000 Hz in freely moving mice using the free Doric Neuroscience studio Software. Demodulated data were further processed by a custom-made software provided by Metofico and based on MATLAB. Briefly, data were downsampled to 1 Hz and artifacts removed using a custom digital filter. Fluorescent increments were normalized to baseline (Δ F/F₀), consisting in the average fluorescence in 1 min window before the introduction of the beetle. To correct movement artifacts, normalized 405 nm signal was subtracted from the normalized 465 nm.

Chemogenetic inhibition

Inactivation of M2 cortex was done using the system of designer receptor exclusively activated by designer drug (DREADD). According to Wang et al. (2020), mice expressing AAV5-hSyn-hM3D(Gi)-mCherry in M2 cortex were intraperitoneally injected \sim 40 min before the BMT with clozapine-N-oxide (CNO, BML-NS105-0025 Enzo) (1.5 mg/kg) or saline as control.

Optogenetic stimulation and multielectrode array (MEA) recordings

Electrophysiological recordings were performed as previously described (Fernández-García et al., 2020; Kim et al., 2020). Briefly, coronal sections of mouse brain were obtained on a vibratome at 350 μ m thickness in ice-cold aCSF oxygenated (95% O₂, 5% CO₂) and then transferred for 15 min to an oxygenated 32°C recovery solution. Slices were then transferred to oxygenated aCSF at room temperature for at least 1 h before recording.

Electrophysiological data were recorded by a blind experimenter using a MEA set-up from Multi Channel Systems MCS. MC Stimulus and MC Rack software from Multi Channel Systems were used for stimulation, recording, and signal processing. We assessed the correct position of the slices on the electrode field using a digital camera during the recording.

Field postsynaptic currents (fPSCs) were recorded in the lateral SC in response to the optogenetic stimulation of M2 cortical afferents with 1 ms 473 nm light pulses of increasing intensities (0.024, 0.033, 0.911, 4.23, 10.2, 19.8, 44.8, and 79.0 mW/mm²), delivered by a 473 nm diode pumped solid state laser (Laserglow). The laser was approximately placed in the surface of the slide at \sim -4.04 posterior to bregma, -1.75 ventral from the skull, and ±1.45 lateral from the middle, corresponding

to lateral SC. Evoked fPSCs responses in the SC were analyzed after trains of 1 ms light stimuli.

Immunohistochemistry

Mice were killed by cervical dislocation and brains were postfixed with 4% PFA for immunohistochemical analysis, dehydrated in a PBS/sucrose gradient [from 15% (48 h postmortem) to 30% (32 h postmortem)] with 0.02% sodium azide, snap frozen and stored at $-20^\circ\text{C}.$ Sections (30 $\mu\text{m})$ were cut on a vibratome (Leica VT1000S) and preserved in PBS with 0.02% sodium azide at 4°C. Anti-GFP (1:500, Invitrogen, #11122, RRID: AB_221569) antibody was used to evaluate expression of AAV-YFP, AAV-GCAMP6f. Free-floating sections were washed in PBS and permeabilized and blocked for 15 min in PBS containing 0.3% Triton X-100 and 3% normal goat serum (Pierce Biotechnology). Sections were washed again in PBS and incubated overnight at 4°C with primary antibodies. Brain slices were washed, incubated for 2 h with the secondary antibody (1:200 goat anti-rabbit Cy3, Jackson ImmunoResearch Laboratories, catalog #111-165-003, RRID:AB_2338000, and mounted on microscope slides using DAPI Fluoromount-G, Southern Biotechnology). Fluorescence images were acquired by an epifluorescence microscope (DMI6000 Leica). No signal was detected in absence of primary antibody. Mice with no GFP expression observed, indicating no expression of AAVs, were discarded.

Statistical analysis

All the results were expressed as mean \pm SEM. Data from individual mouse was represented by single points when possible. GraphPad Prism version 8.0.0. Software was used for statistical analysis. Student's *t* test and the one-way and the two-way ANOVA analysis, followed by Bonferroni *post hoc* test were performed when appropriate, and indicated in Results and/or figure legends. Values of p < 0.05 were considered as statistically significant.

Results

The M2 cortex is functionally disconnected with the SC in HD mice

To investigate M2 cortex connectivity alterations in HD, we used rs-fMRI and analyzed functional connectivity using seed-based analysis between the M2 cortex and 22 automatically identified brain regions covering the left brain hemisphere in ~20-weekold WT and R6/1 mice (Fig. 1). M2 cortex showed the highest functional connectivity with Cg and SC in WT mice. In addition, functional connectivity between the M2 cortex and all the brain regions analyzed was reduced in HD mice compared with WT. Two-way ANOVA reported a significant genotype effect ($F_{(1,22)} = 11.9$; p = 0.002), brain region effect ($F_{(21\,462)} =$ 35.9, p < 0.0001), and region × genotype interaction effect $(F_{(21,462)} = 3.6, p < 0.0001)$. The left M2 cortex of HD mice, compared with WT, showed significantly reduced functional connectivity with left SC (p < 0.0001), but also with left periaqueductal gray (PAG; p = 0.001), thalamus (Thal; p = 0.05), hippocampus (Hip; p = 0.01), RS (p = 0.001), Cg (p = 0.007),



Figure 5. Behavioral responses toward a randomly moving robo-beetle differ between WT and HD mice at symptomatic stages. *A*, Schematic representation of the BMT protocol, including 5 min habituation and 5 min test phases in a rectangular arena. *B*, *C*, During the habituation phase, (*B*) distance traveled and (*C*) rearing time were measured. *D*-*I*, During the test phase, (*D*) total induced responses to the robo-beetle are represented. Percentages of (*E*) escape responses, (*F*) total responses excluding escape, and percentage of (*G*) tolerance, (*H*) approach, and (*I*) avoidance responses when excluding the escape response were analyzed. Each point represents data from an individual mouse. Unpaired Student's *t* test was performed. Data are mean \pm SEM (WT *n* = 12 and HD *n* = 14 mice; 21-week-old mice). **p* < 0.05. ***p* < 0.001.

Som (p = 0.02), and mPFC (p = 0.02), as shown by Bonferroni *post hoc* test (Fig. 1). Thus, our results highlight that M2 cortex functional connectivity deficits in symptomatic HD mice involves several cortical, hippocampal, and non-basal ganglia brain regions, with most prominent changes with the SC.

Structural alterations in M2 cortex projection to SC in HD mice

To evaluate whether functional connectivity deficits observed in M2 cortex-SC circuit were associated to structural alterations, we estimated the M2 cortex-SC tracts from DWI and measured FA as well as MD, AD, and RD diffusivity (Fig. 2). Our data indicate that structural alterations are also present in the M2 cortex-SC tract. We reported significant decreased FA values (Fig. 2B) when comparing HD with WT mice, while MD, AD, and RD remained similar between genotypes (Fig. 2C-E). In addition, we analyzed DTI metrics within the SC to exclude possible regional microstructural alterations (Fig. 3). All the evaluated diffusion metrics measured in the SC were similar between WT and HD mice, suggesting that microstructure alterations in the M2 cortex-SC are related to the projection between both brain regions, but not to the SC alone.



Figure 6. The behavioral response to unexpected flash of light is altered in symptomatic HD mice. *A*, Schematic representation of the task protocol. A mouse will travel through a corridor, and a sudden flash of light is opened when the mouse reaches the center. *B*, Plot of average time to travel until the center (Time 1, t1) and from the center to the end of the corridor (Time 2, t2). Each point represents the average of the 5 trials from an individual mouse. *C*, *D*, Graphs showing the measures of Time 1 (*C*) and Time 2 (*D*) over the 5 trials. Two-way ANOVA with genotype and time or trials as factors was performed and followed by Bonferroni *post hoc* test comparisons. Data are mean \pm SEM (WT *n* = 12 and HD *n* = 14 mice; 21-week-old mice). **p* < 0.05. ****p* < 0.001.

Optogenetic stimulation of M2 cortical axons in the SC evoked similar responses in WT and HD mice

To elucidate whether SC respond to M2 cortex activity in both WT and HD mice, we performed MEA recordings in the deep layers of the lateral part of the of the SC (dlSC) in WT and HD mice, which were previously injected in the M2 cortex with the AAV-CamKII-ChR2-YFP construct (Fig. 4). A normalized input-output assay was created by recording fPSC in the dlSC induced by increasing light intensities. Our results revealed that optical stimulation of M2 cortical axons in the dlSC evoked fPSC

in both WT and HD mice. Two-way ANOVA showed a significant light intensify effect ($F_{(7,84)} = 34.5$, p < 0.0001), although neither genotype effect ($F_{(1,12)} = 3.0$, p = 0.1) nor genotype × light intensity interaction ($F_{(7,84)} = 1.9$, p = 0.08) were found.

Impaired SC-associated behavioral functions in symptomatic HD mice

The SC receives and integrates visual information to control reflex movements, such as stimulus orienting (approach



Figure 7. Behavioral responses toward a randomly moving robo-beetle are altered in both male and female HD mice compared with WT at 12 weeks of age. *A*, *B*, During the habituation phase, (*A*) distance traveled and (*B*) rearing time were measured. *C*-*H*, During the test phase, (*C*) total induced responses to the robo-beetle are represented. Percentages of (*D*) escape responses, (*E*) total responses excluding escape, and percentage of (*F*) tolerance, (*G*) approach, and (*H*) avoidance responses when excluding the escape response were analyzed. Each point represents data from an individual mouse. Two-way ANOVA with genotype and sex as factors was performed. Data are mean \pm SEM (male WT n = 8; male HD n = 10; female WT n = 9; female HD n = 9 mice; 12-week-old mice). *p < 0.05. **p < 0.01.

response) or defensive movements (avoidance or flight) (Wurtz and Albano, 1980; Sparks, 1986; Dean et al., 1989). To examine whether SC-dependent behaviors were affected in HD mice, we selected two tasks known to involve SC function in rodents: BMT (Almada et al., 2018) and the visually induced locomotor behavior (Liang et al., 2015).

In the BMT (Fig. 5), the behavioral responses to a randomly moving robo-beetle are evaluated. During the habituation phase of the test, mice were allowed to explore the arena for 5 min and locomotion and spontaneous exploration, such as rearing time, was significantly decreased in HD mice compared with WT (Fig. 5B,C), as previously described (Pépin et al., 2016). During the testing phase, when mice were confronted with the robo-beetle, HD mice showed a significant decrease in total number of responses to the robo-beetle (Fig. 5D). Then, we evaluated the type of response toward the robo-beetle contact, and we observed that the major response in WT mice was to escape from the robo-beetle, compared with the other evaluated responses, such as tolerance, approach, or avoidance. Interestingly, HD mice show little or non-escape response, as demonstrated by the strongly reduced total percentage of escape response (Fig. 5E). We further evaluated the percentage of responses when excluding escape behavior, as previously described (Almada et al., 2018) and found that number of total responses from HD mice toward the robo-beetle was significantly higher compared with WT when obviating escape responses (Fig. 5F). Of those, HD mice had significantly increased both tolerance (Fig. 5G) and approach responses (Fig. 5H). Finally, although avoidance behavior is the preferred response to the robo-beetle in both genotypes (when excluding escape responses), HD mice showed significantly reduced avoidance responses compared with WT mice (Fig. 51). Together, our data highlight that the choice of response to unexpected threatening beetle differs in HD compared WT mice, suggesting alterations in coupling of visual inputs and thread responding.

Next, we assessed light-triggered locomotor behaviors in response to unexpected stimuli from the upper-visual field

(Liang et al., 2015). We compared the response to a sudden flash of white light (\sim 1 s duration) in the center of a dark corridor between genotypes (Fig. 6). We observed that a sudden light induced a short freezing (arrest behavior) in WT mice (data not shown). Although we did not observe reduction in time to travel through the corridor after the light in WT mice as previously described (Liang et al., 2015), we found this time increased in HD mice. Two-way ANOVA reported a significant genotype effect ($F_{(1,24)} = 11.0$, p = 0.003), Time 1/ Time 2 effect ($F_{(1,24)} = 20.8$, p = 0.0001), and interaction effect $(F_{(1,24)} = 10.9, p = 0.003)$. Bonferroni *post hoc* analysis showed that locomotion was not affected in WT mice by the sudden light (Fig. 6B), whereas it was significantly increased in HD mice and consistent among trials (Fig. 6C,D). On the one hand, time to reach the center of the corridor was affected by the trial number, as shown by two-way ANOVA with a significant trial effect ($F_{(4,85)} = 6.8$, p < 0.0001), but neither genotype effect ($F_{(1,24)} = 4.3, p = 0.05$) nor interaction ($F_{(4,85)} = 6.8$, p = 0.09) (Fig. 6C). On the other hand, time from the center of the corridor to the end was increased in all trials in HD mice, as shown by two-way ANOVA significant genotype effect ($F_{(1,24)}$ = 15.8, p = 0.0006) and interaction ($F_{(4,85)} = 3.8$, p = 0.007), but no trial effect ($F_{(4,85)} = 1.1$, p = 0.3) (Fig. 6D). Therefore, our data further confirm that WT and HD mice generate divergent responses in response to unexpected visual stimuli.

To exclude that gender or motor alterations could confound the present results, the BMT was performed in presymptomatic male and female mice at 12 weeks of age, when no motor coordination symptoms are yet manifested (Canals et al., 2004; Anglada-Huguet et al., 2014) (Fig. 7). Our results confirm that at this stage locomotor activity is not altered, while spontaneous behavior such as rearing is already reduced in HD mice compared with control. Moreover, altered responses toward the randomly moving robo-beetle were similarly present at this stage in both sexes, with reduced number of responses, similar total responses without escape, increased tolerance and reduced avoidance and approach responses in HD mice compared with control WT mice (Fig. 7). In sum, SC-related behaviors are altered in HD mice from presymptomatic stages.

Visual stimuli engage neuronal activity in the M2 cortex of WT but not HD mice

Because neurons from the M2 cortex project to the dISC and are associated to lower visual stimulus (Comoli et al., 2012), we hypothesize that the activity of M2 cortex neurons might contribute to alterations in the BMT observed in HD mice. To prove this, we assessed whether M2 cortex activity correlated with the presence of an unexpected lower-visual stimuli, such as the presence of a robobeetle (Fig. 8). We expressed a GCaMP6f calcium sensor in M2 cortical neurons by injecting an AAV-GCaMP6f construct (Fig. 8A,B) and recorded fluorescence during the 10 min of the BMT (Fig. 8C). Fluorescence signal was stable and similar between genotypes during the habituation phase, and markedly increased straight after introducing the robo-beetle in all WT mice, but not in HD mice (Fig. 8D).

To better quantify these results, we averaged fluorescent data, to obtain 1 min windows, and evaluated genotypes differences over time. Two-way ANOVA analysis reported a significant time effect $(F_{(5,45)} = 4.7, p = 0.002)$, genotype effect $(F_{(1,9)} = 7.4, p = 0.02)$, and time × genotype interaction $(F_{(5,19)} = 3.1, p = 0.02)$ (Fig. 8E). Bonferroni post hoc analysis showed significant differences between groups during the first 3 min of the testing phase. Thus, our data confirm that M2 cortex becomes active in WT mice in the presence of an unexpected lowervisual stimuli. However, this neuronal engagement is lacking in HD mice.

Interestingly, while all WT mice engaged M2 activity, not all of them showed escape responses from the robo-beetle in this experiment (Fig. 9), suggesting that M2 cortical engagement does not exclusively reflect this specific behavioral response. Additionally, and despite the low number of mice, tolerance responses were significantly increased and avoidance responses were significantly reduced, in line with results from Figure 5.

Acute inhibition of M2 cortical neurons in WT mice slightly shifts WT responses in the BMT toward an HD phenotype

To further understand whether the lack of M2 cortex activity is responsible for the HD phenotype in the BMT, we



Figure 8. The presence of visual stimuli activates M2 cortex neurons in WT but not HD mice, shown by fluorescent calcium recordings using fiber photometry *in vivo.* **A**, Schematic representation of fiber-optic cannula implant and representative image of the injection of the AAV-GCaMP6f construct at M2 cortex. **B**, Representative fluorescent image showing DAPI (blue), GCaMP6f expression (green), and fiber-optic cannula implant in the M2 cortex of WT (left) and HD (right) mice. **C**, BMT protocol, including 5 min habituation and 5 min test. **D**, Mean photometric recordings of M2 cortex fluorescent calcium signal 1 min before and 5 min after the introduction of a moving robo-beetle to the arena in WT and symptomatic HD mice. Increases in fluorescent signal are normalized by the respective baseline for each mouse. **E**, Average of normalized fluorescence signal is computed and represented for each minute. Each point represents data from an individual mouse. Two-way ANOVA analysis with genotype and time as factors was performed and followed by Bonferroni *post hoc* test comparisons. Data are mean ± SEM (WT *n* = 6 and HD *n* = 5 mice; 20-week-old mice). **p* < 0.05, ***p* < 0.01, versus WT.



Figure 9. Behavioral responses during the BMT coupled to *in vivo* fiber photometry in WT and HD mice, related to Figure 8. *A*, *B*, During the habituation phase, (*A*) distance traveled and (*B*) rearing time were measured. *C*-*H*, During the test phase, (*C*) total responses to the robo-bettle, (*D*) percentage of escape responses, (*E*) total responses excluding escape, and percentage of (*F*) tolerance, (*G*) approach, and (*H*) avoidance responses when excluding the escape response were analyzed. Each point represents data from an individual mouse. Unpaired Student's *t* test was performed. Data are mean \pm SEM (WT *n* = 6 and HD *n* = 6 mice; 20-week-old mice). ****p* < 0.001.

examined whether acute inhibition of M2 cortical neurons in WT mice could mimic HD responses to a lower visual input. We bilaterally injected mice with AAV5-hSyn-hM3D(Gi)-mCherry in M2 cortex to express the Gi-protein-coupled receptor hM3D and compared the behavioral performance in the BMT ~40 min after an intraperitoneal injection of CNO (1 mg/kg) or saline (Veh) (Fig. 10).

During the habituation phase, both locomotion and exploratory behavior remain similar between WT vehicle or CNO injected mice, while was reduced in HD mice (Fig. 10*B*,*C*). One-way ANOVA revealed significant group effect in both locomotion ($F_{(2,11)} = 10.6$, p = 0.03) and exploratory behavior ($F_{(2,11)} = 13.5$, p = 0.001), with only a significant reduction in both parameters in HD mice compared with WT, as revealed by Bonferroni *post hoc* analysis.

During the subsequent testing phase, the behavioral responses toward the robo-beetle were evaluated. The responses toward the robo-beetle in WT CNO-treated mice were shifted toward an HD phenotype (Fig. 10D-I). In detail, CNO treatment in WT mice increased total number of responses compared with WT vehicle and got closer to HD vehicle levels (Fig. 10D, one-way ANOVA group effect ($F_{(2,11)} = 7.9, p = 0.008$)). Escape behavior was mostly absent in this experiment and similar in all groups (Fig. 10E, one-way ANOVA group effect $(F_{(2,10)} = 0.8, p = 0.4)$). Also, WT CNO increased the total number of responses without escape behavior (Fig. 10F, oneway ANOVA group effect ($F_{(2,11)} = 8.8, p = 0.005$)), showed a tendency to an increase in tolerance (Fig. 10G, one-way ANOVA group effect ($F_{(2,11)} = 11.5$, p = 0.002)), similar approach (Fig. 10*H*, one-way ANOVA group effect ($F_{(2,11)} = 1.0, p = 0.4$)) and reduced avoidance responses (Fig. 10I, one-way ANOVA group effect $(F_{(2,11)} = 8.6, p = 0.006))$ compared with WT veh and shifted toward the HD phenotype. Bonferroni post hoc analysis further confirmed that inhibition of M2 cortical neurons in WT mice induced a partial HD phenotype, with no differences either to WT Veh or HD veh in any of the responses analyzed, except for the tolerance behavior which was still different to HD veh.

In sum, despite the low number of mice used, acute M2 cortex inhibition in WT mice was able to shift behavioral responses toward an HD phenotype, suggesting that neuronal activity in the M2 cortex is key to select the type of responses on unexpected visual stimuli, such as a randomly moving robo-beetle.

Discussion

Cortical premotor areas are prominently affected in HD many years before the onset of motor symptoms. These alterations have been widely correlated to cortico-striatal disconnection and associated to motor deficits, which are severely manifested in HD. Here, we thoroughly mapped M2 cortex connectivity in HD mice and described prominent structural and functional defects, particularly in the M2 cortex projection to the SC. Moreover, our data reveal profound alterations in SC-dependent behavioral functions in HD mice, which appear before motor alterations. Moreover, we showed that M2 cortex activity is engaged in the presence of a visual stimuli in WT but not in HD mice, and that acute M2 cortex inhibition using DREADDS promotes an HDlike phenotype in the BMT in WT mice, suggesting that aberrant activation of the M2 cortex might contribute to the observed behavioral alterations in HD mice.

M2 cortex pyramidal neurons send projections to many cortical and subcortical brain nuclei, including the SC, as previously described (Zhang et al., 2016; Economo et al., 2018; Gerfen et al., 2018; Lin et al., 2018). Notably, fMRI shows sever functional connectivity deficits between the M2 cortex and several cortices, hippocampus, thalamus, PAG, and SC in symptomatic HD mice, which differs from the neuroanatomical projections. Because functional connectivity is defined as the temporal correlation between spatially remote neurophysiological events (Fingelkurts et al., 2005), the strong correlation of activity between M2 cortex and hippocampus, thalamus, and PAG might be indirect though additional bran circuits. Nevertheless, the M2 cortex represents one of the major inputs to the SC. Further, deficits in functional connectivity are accompanied by structural M2 cortex–SC circuit alterations, seen by a decrease



Figure 10. Chemogenetic inhibition of M2 cortical neurons shifts behavioral responses to the robo-beetle in WT mice toward an HD phenotype in the BMT. *A*, Representative fluorescent image showing M2 cortical neurons expressing AAV5-syn-hdM3D(gi)-mCherry chemogenetic viral constructs (red) and DAPI (blue). *B*, *C*, During the habituation phase (*B*) distance traveled and (*C*) rearing time were measured. *D*-*I*, During the test phase, (*D*) total responses to the robo-beetle were measured. *E*, Percentages of escape response. *F*, Total responses to the robo-beetle excluding escape. *G*, Percentage of tolerance, (*H*) approach and (*J*) avoidance responses from total responses excluding the escape response. Each point represents data from an individual mouse. One-way ANOVA followed by Bonferroni *post hoc* analysis was performed. Data are mean ± SEM (WT-Veh n = 5, WT-CNO n = 4, and HD n = 5 mice; 20-week-old mice). *p < 0.05, **p < 0.01, versus WT-Veh. *p < 0.05, ##p < 0.01, versus WT-Veh. *p < 0.05, ##p < 0.01, versus WT-Veh.

in DWI metrics. Together, we demonstrate that long-range pyramidal tract neurons from M2 cortex to SC are severely affected both structurally and functional, which contrasts with the idea that intratelencephalic neurons are more affected in HD (Gatto et al., 2015). Moreover, SC structure seems preserved in HD conditions, as shown by DWI measurements in the SC and electrophysiological responses if M2 axons in the SC using MEA, suggesting that alterations in M2 cortex–SC functions are related to M2 cortex inputs rather than deficits in the SC structure.

Consistently, alterations in M2 cortex function might be a common underlying mechanism in diverse HD symptoms. M2 cortex activation has been involved in many naturalistic behaviors, such as rearing, grasping, eating, grooming, etc., as seen by calcium imaging experiments using miniature microscopes (Tombaz et al., 2020) and motor learning processes, seen by twophoton imaging of Arc-GFP expression (Cao et al., 2015). All these behaviors are well known to be altered in HD mouse models (Pépin et al., 2016; Fernández-García et al., 2020). In this line, we demonstrated that aberrant M2 cortex activity contributes to the SC-related behaviors in HD, such as the BMT, associated to visual perception of a thread and dependent on SC function (Comoli et al., 2012; Almada et al., 2018). In detail, HD mice showed reduced escape and avoidance responses, while increased tolerance responses to the robo-beetle. Moreover, M2 cortex activity was engaged in the presence of the robo-beetle in WT but not HD mice, and M2 cortex inhibition using DREADDs mildly shifted WT mice responses toward an HD phenotype in the BMT. Although the low number of mice used in the latter experiment could underestimate the effects of chemogenetic stimulation, our data suggests that M2 cortex activity contributes to the selection of responses on visual stimuli; however, additional SC circuits may also contribute to the HD phenotype.

In this regard, M2 cortex displays the highest functional connectivity with the Cg, which in turn is the main cortical afferent to the medial portion of the SC (mSC) (Savage et al., 2017). Information flows from M2 cortex and Cg to the SC in functionally segregated loops and then goes back to basal ganglia circuitry through distinct thalamic targets to modulate action selection (McHaffie et al., 2005). In detail, M2 cortex-dlSC pathway function is related to approach and appetitive stimuli normally found in the lower visual field (Dean et al., 1986; Sahibzada et al., 1986), while Cg-mSC function is more related to escape behaviors and movements in the upper visual field (e.g., predators) (Comoli et al., 2012; Savage et al., 2017). Surprisingly, we found that the number of escape responses in WT mice were very variable among different mice batches, possibly related to differential anxiety levels between them (Heinz et al., 2017), while HD mice did few or no escape responses in any of them. This significant difference observed in evoked escape responses between genotypes suggest that the Cg cortex-mSC and/or the SC-PAG pathways may be also affected in HD pathology. In addition, M2 cortex activity was engaged in all WT mice on the presence of the robo-beetle, regardless of the presence of escape responses, supporting the idea that escape responses are not dependent on M2 cortex activity. Consequently, the presence of alterations in all the diverse behavioral responses in HD mice suggests that inputs to both the medial and lateral parts of the SC might be affected. Accordingly, Cg is known to be altered in HD patients at early stages (Hobbs et al., 2011), and cell loss in the motor and Cg correlates with symptomatology in HD (Thu et al., 2010). Therefore, we hypothesize that alterations in Cg-SC might also have a role in HD.

In parallel, M2 cortex-PAG circuit is also functionally altered in HD mice, as shown by the fMRI data. The PAG is one of the major outputs of the SC and is responsible for defining the threshold to compute escape responses (Evans et al., 2018; Branco and Redgrave, 2020), which were almost absent in HD conditions. Interestingly, increased GABAergic tone from the ventral lateral geniculate nucleus (Fratzl et al., 2021) or the substantia nigra reticulata (Almada et al., 2018) reduces escape responses in the BMT, suggesting that increased GABAergic activity in the SC could lead to the altered behavioral responses in HD mice. Further, SC-PAG circuit is shaped by multimodal sensory information and learning experiences, which are constantly updated throughout the life of the animal (Evans et al., 2019). Because M2 cortex sends only few direct projections to the PAG (Savage et al., 2017), we hypothesize that M2 cortex might modulate PAG function indirectly through the modulation of SC activity, or alternatively, by additional inputs to the SC and PAG, such as the Cg.

The SC is involved in the control of orienting gaze shifts, also known as saccadic movements (Sparks, 1986; Munoz and Guitton, 1991). Therefore, the observed alterations in SC functions in HD mice are consistent with the abnormal control of saccadic eye movements present in HD patients from presymptomatic stages (Patel et al., 2012; Srivastava et al., 2018). Indeed, oculomotor functions and locomotion share many underlying neuronal circuits (Srivastava et al., 2018) and, in several movement disorders, oculomotor deficits precede motor symptoms (Termsarasab et al., 2015). Whereas direct alterations of SC function in HD patients have not been explored, studies in Parkinson disease (PD) show that responses to luminescence (measured by fMRI) in the SC are abnormal in *de novo* patients compared with controls (Moro et al., 2020). These alterations in the SC are linked to deficits in the coordination of action and perception (Pretegiani et al., 2019). Notably, a recent study with PD patients show that repetitive transcranial magnetic stimulation in the motor cortex improved anti-saccade success rate and postural instability gait difficulty (Okada et al., 2021). In this line, optogenetic stimulation of the M2 cortex reverts motor dysfunction in a mouse model of PD (Magno et al., 2019). Thus, further understanding on the complex circuitry between cortical, visual, and motor domains could help to design new treatment opportunities based on circuit restoration in movement disorders.

In conclusion, our data provide compelling evidence of the involvement of M2 cortex circuitry alterations to HD pathophysiology, beyond basal ganglia. Particularly, we highlight the alterations in M2 cortex functional connectivity and particularly the lack of M2 cortex engagement as contributor to the SC-dependent symptoms observed in HD mice. Unraveling the myriad of circuit alterations and functions from M2 cortex other than cortico-striatal provides valuable insights to understand HD symptoms, which might be also relevant for other movement disorders, such as PD. Moreover, further understanding and targeting SC circuitry may additionally provide novel therapeutic opportunities aiming to delay the onset and severity of symptoms in movement disorders.

References

- Almada RC, Genewsky AJ, Heinz DE, Kaplick PM, Coimbra NC, Wotjak CT (2018) Stimulation of the nigrotectal pathway at the level of the superior colliculus reduces threat recognition and causes a shift from avoidance to approach behavior. Front Neural Circuits 12:36–39.
- Anglada-Huguet M, Xifró X, Giralt A, Zamora-Moratalla A, Martín ED, Alberch J (2014) Prostaglandin E2 EP1 receptor antagonist improves

motor deficits and rescues memory decline in *R6/1* mouse model of Huntington's disease. Mol Neurobiol 49:784–795.

- Avants BB, Epstein CL, Grossman M, Gee JC (2008) Symmetric diffeomorphic image registration with cross-correlation: evaluating automated labeling of elderly and neurodegenerative brain. Med Image Anal 12:26– 41.
- Barthas F, Kwan AC (2017) Secondary motor cortex: where 'sensory' meets 'motor' in the rodent frontal cortex. Trends Neurosci 40:181–193.
- Branco T, Redgrave P (2020) The neural basis of escape behavior in vertebrates. Annu Rev Neurosci 43:417–439.
- Canals JM, Pineda JR, Torres-Peraza JF, Bosch M, Martín-Ibañez R, Muñoz MT, Mengod G, Ernfors P, Alberch J (2004) Brain-derived neurotrophic factor regulates the onset and severity of motor dysfunction associated with enkephalinergic neuronal degeneration in Huntington's disease. J Neurosci 24:7727–7739.
- Cao VY, Ye Y, Mastwal S, Ren M, Coon M, Liu Q, Costa RM, Wang KH (2015) Motor learning consolidates Arc-expressing neuronal ensembles in secondary motor cortex. Neuron 86:1385–1392.
- Cepeda C, Wu N, André VM, Cummings DM, Levine MS (2007) The corticostriatal pathway in Huntington's disease. Prog Neurobiol 81:253–271.
- Comoli E, Favaro PD, Vautrelle N, Leriche M, Overton PG, Redgrave P (2012) Segregated anatomical input to sub-regions of the rodent superior colliculus associated with approach and defense. Front Neuroanat 6:9–19.
- Costa RM, Cohen D, Nicolelis MA (2004) Differential corticostriatal plasticity during fast and slow motor skill learning in mice. Curr Biol 14:1124– 1134.
- Coupe P, Yger P, Prima S, Hellier P, Kervrann C, Barillot C (2008) An optimized blockwise nonlocal means denoising filter for 3-D magnetic resonance images. IEEE Trans Med Imaging 27:425–441.
- Creus-Muncunill J, Badillos-Rodríguez R, Garcia-Forn M, Masana M, Barriga GG, Guisado-Corcoll A, Alberch J, Malagelada C, Delgado-García JM, Gruart A, Pérez-Navarro E (2019) Increased translation as a novel pathogenic mechanism in Huntington's disease. Brain 142:3158– 3175.
- Dean P, Redgrave P, Sahibzada N, Tsuji K (1986) Head and body movements produced by electrical stimulation of superior colliculus in rats: effects of interruption of crossed tectoreticulospinal pathway. Neuroscience 19:367–380.
- Dean P, Redgrave P, Westby GW (1989) Event or emergency? Two response systems in the mammalian superior colliculus. Trends Neurosci 12:137– 147.
- Dogan I, Eickhoff CR, Fox PT, Laird AR, Schulz JB, Eickhoff SB, Reetz K (2015) Functional connectivity modeling of consistent cortico-striatal degeneration in Huntington's disease. Neuroimage Clin 7:640–652.
- Duan CA, Pan Y, Ma G, Zhou T, Zhang S, Xu N (2021) A cortico-collicular pathway for motor planning in a memory-dependent perceptual decision task. Nat Commun 12:2727.
- Economo MN, Viswanathan S, Tasic B, Bas E, Winnubst J, Menon V, Graybuck LT, Nguyen TN, Smith KA, Yao Z, Wang L, Gerfen CR, Chandrashekar J, Zeng H, Looger LL, Svoboda K (2018) Distinct descending motor cortex pathways and their roles in movement. Nature 563:79–84.
- Estevez-Fraga C, Scahill R, Rees G, Tabrizi SJ, Gregory S (2021) Diffusion imaging in Huntington's disease: comprehensive review. J Neurol Neurosurg Psychiatry 92:62–69.
- Evans DA, Stempel AV, Vale R, Ruehle S, Lefler Y, Branco T (2018) A synaptic threshold mechanism for computing escape decisions. Nature 558:590–594.
- Evans DA, Stempel AV, Vale R, Branco T (2019) Cognitive control of escape behaviour. Trends Cogn Sci 23:334–348.
- Fernández-García S, Conde-Berriozabal S, García-García E, Gort-Paniello C, Bernal-Casas D, García-Díaz Barriga G, López-Gil J, Muñoz-Moreno E, Soria G, Campa L, Artigas F, Rodríguez MJ, Alberch J, Masana M (2020) M2 cortex-dorsolateral striatum stimulation reverses motor symptoms and synaptic deficits in Huntington's disease. Elife 9:1–24.
- Fingelkurts AA, Fingelkurts AA, K\u00e4hkönen S (2005) Functional connectivity in the brain: is it an elusive concept? Neurosci Biobehav Rev 28:827–836.
- Fratzl A, Koltchev AM, Vissers N, Tan YL, Marques-Smith A, Stempel AV, Branco T, Hofer SB (2021) Flexible inhibitory control of visually evoked defensive behavior by the ventral lateral geniculate nucleus. Neuron 109:3810–3822.e9.

- Gatto RG, Chu Y, Ye AQ, Price SD, Tavassoli E, Buenaventura A, Brady ST, Magin RL, Kordower JH, Morfini GA (2015) Analysis of YFP(J16)-R6/2 reporter mice and postmortem brains reveals early pathology and increased vulnerability of callosal axons in Huntington's disease. Hum Mol Genet 24:5285–5298.
- Gerfen CR, Economo MN, Chandrashekar J (2018) Long distance projections of cortical pyramidal neurons. J Neurosci Res 96:1467–1475.
- Huntington's Disease Collaborative Research Group (1993) A novel gene containing a trinucleotide repeat that is expanded and unstable on Huntington's disease chromosomes. Cell 72:971–983.
- Heinz DE, Genewsky A, Wotjak CT (2017) Enhanced anandamide signaling reduces flight behavior elicited by an approaching robo-beetle. Neuropharmacology 126:233–241.
- Hintiryan H, Foster NN, Bowman I, Bay M, Song MY, Gou L, Yamashita S, Bienkowski MS, Zingg B, Zhu M, Yang XW, Shih JC, Toga AW, Dong HW (2016) The mouse cortico-striatal projectome. Nat Neurosci 19:1100–1114.
- Hobbs NZ, Pedrick AV, Say MJ, Frost C, Dar Santos R, Coleman A, Sturrock A, Craufurd D, Stout JC, Leavitt BR, Barnes J, Tabrizi SJ, Scahill RI (2011) The structural involvement of the cingulate cortex in premanifest and early Huntington's disease. Mov Disord 26:1684–1690.
- Jenkinson M, Beckmann CF, Behrens TE, Woolrich MW, Smith SM (2012) Review FSL. Neuroimage 62:782–790.
- Johnson EB, Ziegler G, Penny W, Rees G, Tabrizi SJ, Scahill RI, Gregory S (2021) Dynamics of cortical degeneration over a decade in Huntington's disease. Biol Psychiatry 89:807–816.
- Kim A, García-García E, Straccia M, Comella-Bolla A, Miguez A, Masana M, Alberch J, Canals JM, Rodríguez MJ (2020) Reduced fractalkine levels lead to striatal synaptic plasticity deficits in Huntington's disease. Front Cell Neurosci 14:163.
- Kupferschmidt DA, Juczewski K, Cui G, Johnson KA, Lovinger DM (2017) Parallel, but dissociable, processing in discrete corticostriatal inputs encodes skill learning. Neuron 96:476–489.e5.
- Liang F, Xiong XR, Zingg B, Ji X, Zhang LI, Tao HW (2015) Sensory cortical control of a visually induced arrest behavior via corticotectal projections. Neuron 86:755–767.
- Lin HM, Kuang JX, Sun P, Li N, Lv X, Zhang YH (2018) Reconstruction of intratelencephalic neurons in the mouse secondary motor cortex reveals the diverse projection patterns of single neurons. Front Neuroanat 12:86.
- Ma Y, Smith D, Hof PR, Foerster B, Hamilton S, Blackband SJ, Yu M, Benveniste H (2008) In vivo 3D digital atlas database of the adult C57BL/6J mouse brain by magnetic resonance microscopy. Front Neuroanat 2:1–10.
- Magno LA, Tenza-Ferrer H, Collodetti M, Aguiar MF, Rodrigues AP, da Silva RS, do Prado Silva J, Nicolau NF, Rosa DV, Birbrair A, Miranda DM, Romano-Silva MA (2019) Optogenetic stimulation of the M2 cortex reverts motor dysfunction in a mouse model of Parkinson's disease. J Neurosci 39:3234–3248.
- McHaffie JG, Stanford TR, Stein BE, Coizet V, Redgrave P (2005) Subcortical loops through the basal ganglia. Trends Neurosci 28:401–407.
- Moro E, Bellot E, Meoni S, Pelissier P, Hera R, Dojat M, Coizet V, Superior Colliculus Study Group (2020) Visual dysfunction of the superior colliculus in de novo Parkinsonian patients. Ann Neurol 87:533–546.
- Munoz DP, Guitton D (1991) Control of orienting gaze shifts by the tectoreticulospinal system in the head-free cat: II. Sustained discharges during motor preparation and fixation. J Neurophysiol 66:1624–1641.
- Okada K, Takahira M, Mano T, Uga T, Konaka K, Hosomi K, Saitoh Y (2021) Concomitant improvement in anti-saccade success rate and

postural instability gait difficulty after rTMS treatment for Parkinson's disease. Sci Rep 11:2472.

- Patel SS, Jankovic J, Hood AJ, Jeter CB, Sereno AB (2012) Reflexive and volitional saccades: biomarkers of Huntington disease severity and progression. J Neurol Sci 313:35–41.
- Pépin J, Francelle L, Carrillo-de Sauvage MA, de Longprez L, Gipchtein P, Cambon K, Valette J, Brouillet E, Flament J (2016) In vivo imaging of brain glutamate defects in a knock-in mouse model of Huntington's disease. Neuroimage 139:53–64.
- Pretegiani E, Vanegas-Arroyave N, FitzGibbon EJ, Hallett M, Optican LM (2019) Evidence from Parkinson's disease that the superior colliculus couples action and perception. Mov Disord 34:1680–1689.
- Sahibzada N, Dean P, Redgrave P (1986) Movements resembling orientation or avoidance elicited by electrical stimulation of the superior colliculus in rats. J Neurosci 6:723–733.
- Savage MA, McQuade R, Thiele A (2017) Segregated fronto-cortical and midbrain connections in the mouse and their relation to approach and avoidance orienting behaviors. J Comp Neurol 525:1980–1999.
- Shaffer JJ, Ghayoor A, Long JD, Kim RE, Lourens S, O'Donnell LJ, Westin CF, Rathi Y, Magnotta V, Paulsen JS, Johnson HJ (2017) Longitudinal diffusion changes in prodromal and early HD: evidence of white-matter tract deterioration. Hum Brain Mapp 38:1460–1477.
- Sparks DL (1986) Translation of sensory signals into commands for control of saccadic eye movements: role of primate superior colliculus. Physiol Rev 66:118–171.
- Srivastava A, Ahmad OF, Pacia CP, Hallett M, Lungu C (2018) The relationship between saccades and locomotion. J Mov Disord 11:93–106.
- Sul JH, Jo S, Lee D, Jung MW (2011) Role of rodent secondary motor cortex in value-based action selection. Nat Neurosci 14:1202–1208.
- Termsarasab P, Thammongkolchai T, Rucker JC, Frucht SJ (2015) The diagnostic value of saccades in movement disorder patients: a practical guide and review. J Clin Mov Disord 2:1–10.
- Thu DC, Oorschot DE, Tippett LJ, Nana AL, Hogg VM, Synek BJ, Luthi-Carter R, Waldvogel HJ, Faull RL (2010) Cell loss in the motor and cingulate cortex correlates with symptomatology in Huntington's disease. Brain 133:1094–1110.
- Tombaz T, Dunn BA, Hovde K, Cubero RJ, Mimica B, Mamidanna P, Roudi Y, Whitlock JR (2020) Action representation in the mouse parieto-frontal network. Sci Rep 10:1–14.
- Tustison NJ, Avants BB, Cook PA, Zheng Y, Egan A, Yushkevich PA, Gee JC (2010) N4ITK: improved N3 bias correction. IEEE Trans Med Imaging 29:1310–1320.
- Unschuld PG, Joel SE, Liu X, Shanahan M, Margolis RL, Biglan KM, Bassett SS, Schretlen DJ, Redgrave GW, van Zijl PC, Pekar JJ, Ross CA (2012) Impaired cortico-striatal functional connectivity in prodromal Huntington's disease. Neurosci Lett 514:204–209.
- Veldman MB, Yang XW (2018) Molecular insights into cortico-striatal miscommunications in Huntington's disease. Curr Opin Neurobiol 48:79–89.
- Wang TY, Liu J, Yao H (2020) Control of adaptive action selection by secondary motor cortex during flexible visual categorization. Elife 9:1–26.
- Wurtz RH, Albano JE (1980) Visual-motor function of the primate superior colliculus. Annu Rev Neurosci 3:189–226.
- Yang JH, Kwan AC (2021) Secondary motor cortex: broadcasting and biasing animal's decisions through long-range circuits. Int Rev Neurobiol 158:443–470.
- Zhang S, Xu M, Chang WC, Ma C, Hoang Do JP, Jeong D, Lei T, Fan JL, Dan Y (2016) Organization of long-range inputs and outputs of frontal cortex for top-down control. Nat Neurosci 19:1733–1742.